%~MoulinÃ© par MaN_auto v.0.33.8 (9c3f6e85) 2025-04-16 16:48:55
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{GÃ©omÃ©trie algÃ©brique}
\TopicEN{Algebraic geometry}

\providecommand\noopsort[1]{}

\addbibresource{CRMATH_Cesnavicius_20240033.bib}

\usepackage{mathtools}
\usepackage[english,capitalize,nameinlink]{cleveref}
\usepackage[all]{xy}

\DeclareMathOperator{\Aut}{Aut}
\DeclareMathOperator{\Bun}{Bun}
\DeclareMathOperator{\GL}{GL}
\DeclareMathOperator{\Gr}{Gr}
\DeclareMathOperator{\Ker}{Ker}
\DeclareMathOperator{\Res}{Res}
\DeclareMathOperator{\Spec}{Spec}

\newcommand{\bA}{\mathbb{A}}
\newcommand{\bC}{\mathbb{C}}
\newcommand{\bG}{\mathbb{G}}
\newcommand{\bP}{\mathbb{P}}
\newcommand{\bQ}{\mathbb{Q}}
\newcommand{\bZ}{\mathbb{Z}}

\newcommand{\bbB}{\mathbf{B}}

\newcommand{\cE}{\mathcal{E}}
\newcommand{\cG}{\mathcal{G}}
\newcommand{\cP}{\mathcal{P}}
\newcommand{\cX}{\mathcal{X}}
\newcommand{\sO}{\mathcal{O}}

\newcommand{\fp}{\mathfrak{p}}

\newcommand{\ad}{\mathrm{ad}}
\newcommand{\alg}{\mathrm{alg}}
\newcommand{\dR}{{\mathrm{dR}}}
\newcommand{\et}{\mathrm{\acute{e}t}}
\newcommand{\fpqc}{\mathrm{fpqc}}
\newcommand{\gp}{{\mathrm{gp}}}
\newcommand{\Zar}{\mathrm{Zar}}
\newcommand{\new}{\mathrm{new}}
\newcommand{\Hensel}{\mathrm{h}}

\newcommand{\tensor}{\otimes}

\newcommand{\q}{\quad}
\newcommand{\qq}{\quad\quad}
\newcommand{\qqq}{\quad\quad\quad}
\newcommand{\qqqq}{\quad\quad\quad\quad}

\newenvironment{swapenonce}[2][remark]{\swapnumbers \begin{enonce}[#1]{#2}}{\end{enonce}}

%%%---------------------------------------------------------------------------

%%% ARROWS

%%% \to macro
\renewcommand*{\to}{\mathchoice{\longrightarrow}{\rightarrow}{\rightarrow}{\rightarrow}}
%% Remplacer \xrightarrow{#1} par \overset{#1}{\longrightarrow}. 

%%% \mapsto macro
\let\oldmapsto\mapsto
\renewcommand*{\mapsto}{\mathchoice{\longmapsto}{\oldmapsto}{\oldmapsto}{\oldmapsto}}

%%% \longhookrightarrow macro
\newcommand{\longhookrightarrow}{\!\lhook\joinrel\longrightarrow}

%%% \longrightrightarrows macro
\newcommand{\longrightrightarrows}{\overset{\raisebox{-.17ex}{$\displaystyle{\longrightarrow}$}}{\smash{\raisebox{-.3ex}{$\longrightarrow$}}}}

%%% Convergence
%\newcommand\dconverge[2][p \to + \infty]{\xrightarrow[#1]{#2}}
%\newcommand\tconverge[1][p \to + \infty]{\underset{#1}{\longrightarrow}}

\newcommand{\ra}{\rightarrow}
\newcommand{\hra}{\hookrightarrow}
\newcommand{\isomto}{\overset{\!\!\raisebox{.3ex}{$\scriptstyle{\sim}$}}{\smash{\to}}}

%%%---------------------------------------------------------------------------

%%% BAR & OVERLINE

%%% Better widebar macro
\makeatletter
\let\save@mathaccent\mathaccent
\newcommand*\if@single[3]{%
  \setbox0\hbox{${\mathaccent"0362{#1}}^H$}%
  \setbox2\hbox{${\mathaccent"0362{\kern0pt#1}}^H$}%
  \ifdim\ht0=\ht2 #3\else #2\fi
  }
%The bar will be moved to the right by a half of \macc@kerna, which is computed by amsmath:
\newcommand*\rel@kern[1]{\kern#1\dimexpr\macc@kerna}
%If there's a superscript following the bar, then no negative kern may follow the bar;
%an additional {} makes sure that the superscript is high enough in this case:
\newcommand*\widebar[1]{\@ifnextchar^{{\wide@bar{#1}{0}}}{\wide@bar{#1}{1}}}
%Use a separate algorithm for single symbols:
\newcommand*\wide@bar[2]{\if@single{#1}{\wide@bar@{#1}{#2}{1}}{\wide@bar@{#1}{#2}{2}}}
\newcommand*\wide@bar@[3]{%
  \begingroup
  \def\mathaccent##1##2{%
%Enable nesting of accents:
    \let\mathaccent\save@mathaccent
%If there's more than a single symbol, use the first character instead (see below):
    \if#32 \let\macc@nucleus\first@char \fi
%Determine the italic correction:
    \setbox\z@\hbox{$\macc@style{\macc@nucleus}_{}$}%
    \setbox\tw@\hbox{$\macc@style{\macc@nucleus}{}_{}$}%
    \dimen@\wd\tw@
    \advance\dimen@-\wd\z@
%Now \dimen@ is the italic correction of the symbol.
    \divide\dimen@ 3
    \@tempdima\wd\tw@
    \advance\@tempdima-\scriptspace
%Now \@tempdima is the width of the symbol.
    \divide\@tempdima 10
    \advance\dimen@-\@tempdima
%Now \dimen@ = (italic correction / 3) - (Breite / 10)
    \ifdim\dimen@>\z@ \dimen@0pt\fi
%The bar will be shortened in the case \dimen@<0 !
    \rel@kern{0.6}\kern-\dimen@
    \if#31
      \overline{\rel@kern{-0.6}\kern\dimen@\macc@nucleus\rel@kern{0.4}\kern\dimen@}%
      \advance\dimen@0.4\dimexpr\macc@kerna
%Place the combined final kern (-\dimen@) if it is >0 or if a superscript follows:
      \let\final@kern#2%
      \ifdim\dimen@<\z@ \let\final@kern1\fi
      \if\final@kern1 \kern-\dimen@\fi
    \else
      \overline{\rel@kern{-0.6}\kern\dimen@#1}%
    \fi
  }%
  \macc@depth\@ne
  \let\math@bgroup\@empty \let\math@egroup\macc@set@skewchar
  \mathsurround\z@ \frozen@everymath{\mathgroup\macc@group\relax}%
  \macc@set@skewchar\relax
  \let\mathaccentV\macc@nested@a
%The following initialises \macc@kerna and calls \mathaccent:
  \if#31
    \macc@nested@a\relax111{#1}%
  \else
%If the argument consists of more than one symbol, and if the first token is
%a letter, use that letter for the computations:
    \def\gobble@till@marker##1\endmarker{}%
    \futurelet\first@char\gobble@till@marker#1\endmarker
    \ifcat\noexpand\first@char A\else
      \def\first@char{}%
    \fi
    \macc@nested@a\relax111{\first@char}%
  \fi
  \endgroup
}
\makeatother

\let\oldbar\bar
\renewcommand*{\bar}[1]{{\mathchoice{\widebar{#1}}{\widebar{#1}}{\widebar{#1}}{\oldbar{#1}}}}

% /!\ Il y a parfois des soucis avec cette macro lorsqu'il y a des caractÃ¨res en minuscule en subscript (et dans les captions). Dans ce cas, faire avec \overline et des espaces manuels.

%%% Double bar
% \overline{\overline{\!a}}

%\newcommand\bx{\,\overline{\!x}}
%\newcommand\brho{\,\overline{\!\rho}}
%better widebar a apparemment des soucis avec les \subcaptions pour les CR :shrug:

%%%---------------------------------------------------------------------------

%%% PACKAGE CLEVEREF

%\usepackage[english,capitalize,nameinlink]{cleveref}
\newcommand*\claimref[1]{\hyperref[#1]{Claim~\ref*{#1}}}
\newcommand*\conjref[1]{\hyperref[#1]{Conjecture~\ref*{#1}}}
\newcommand*\cororef[1]{\hyperref[#1]{Corollary~\ref*{#1}}}
\newcommand*\defiref[1]{\hyperref[#1]{Definition~\ref*{#1}}}
\newcommand*\figureref[1]{\hyperref[#1]{Figure~\ref*{#1}}}
\newcommand*\lemmref[1]{\hyperref[#1]{Lemma~\ref*{#1}}}
\newcommand*\propref[1]{\hyperref[#1]{Proposition~\ref*{#1}}}
\newcommand*\subsectionref[1]{\hyperref[#1]{Section~\ref*{#1}}}
\newcommand*\theoref[1]{\hyperref[#1]{Theorem~\ref*{#1}}}

%%%---------------------------------------------------------------------------

%%% DELIMITERS

%%\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
%\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
%\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
%\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
%\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

%%% VERTICAL BAR

%%% ''Evaluated at'' macro
\newcommand\restr[2]{{% we make the whole thing an ordinary symbol
 \left.\kern-\nulldelimiterspace % automatically resize the bar with \right
 #1 % the function
 \vphantom{\big|} % pretend it's a little taller at normal size
 \right|_{#2} % this is the delimiter
 }}
%%% As a subscript: \phi_{F_f\rvert_{K_\ell}} [NO : \phi_{\left.h\right|_{K_\ell}}].
%%% In textstyle : \bigr\rvert_{}.

%%%% ''Such that'' macro
%\newcommand*{\st}{\;\vert\;}
%\newcommand*{\bigst}{\;\big\vert\;}
%\newcommand*{\Bigst}{\;\Big\vert\;}
%\newcommand*{\biggst}{\;\bigg\vert\;}
%\newcommand*{\Biggst}{\;\Bigg\vert\;}
%% Scaling version (with \left and \right delimiters): \,\middle\vert\,

%%% Divides
% \vert

%%% Separator (non-scaling)
% \mid

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses
% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=12cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\begin{DefTralics}
\DeclareMathOperator{\Bun}{Bun}
\DeclareMathOperator{\GL}{GL}
\DeclareMathOperator{\Gr}{Gr}
\newcommand{\bC}{\mathbb{C}}
\newcommand{\bP}{\mathbb{P}}
\end{DefTralics}

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\newcommand*{\relabel}{\renewcommand{\labelenumi}{(\theenumi)}}
\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}\relabel}
\newcommand*{\Romanenumi}{\renewcommand*{\theenumi}{\Roman{enumi}}\relabel}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}\relabel}
\newcommand*{\Alphenumi}{\renewcommand*{\theenumi}{\Alph{enumi}}\relabel}
\let\oldtilde\tilde
\renewcommand*{\tilde}[1]{\mathchoice{\widetilde{#1}}{\widetilde{#1}}{\oldtilde{#1}}{\oldtilde{#1}}}
\let\oldhat\hat
\renewcommand*{\hat}[1]{\mathchoice{\widehat{#1}}{\widehat{#1}}{\oldhat{#1}}{\oldhat{#1}}}

\title{The affine Grassmannian as a presheaf quotient}
\alttitle{La grassmannienne affine comme quotient de prÃ©faisceaux}

\author{\firstname{K{\k e}stutis} \lastname{{\v C}esnavi{\v c}ius}}
\address{CNRS, Universit\'{e} Paris-Saclay, Laboratoire de math\'{e}matiques d'Orsay, F-91405, Orsay, France}
\email{kestutis@math.u-psud.fr}

\subjclass{14L15, 14M17}

\keywords{\kwd{Affine Grassmannian} \kwd{loop group} \kwd{reductive group} \kwd{torsor}}
\altkeywords{\kwd{Grassmannienne affine} \kwd{groupe de lacets} \kwd{groupe rÃ©ductif} \kwd{torseur}}

\thanks{This project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No.~851146). This project is based upon work supported by the National Science Foundation under grant No.~DMS-1926686}
\CDRGrant[European Research Council (ERC)]{851146}
\CDRGrant[National Science Foundation]{DMS-1926686}

\begin{abstract}
For a reductive group $G$ over a ring $A$, its affine Grassmannian $\Gr_G$ plays important roles in a wide range of subjects and is typically defined as the \'etale sheafification of the presheaf quotient $LG/L^+G$ of the loop group $LG$ by its positive loop subgroup $L^+G$. We show that the Zariski sheafification gives the same result. Moreover, for totally isotropic $G$ (for instance, for quasi-split $G$), we show that no sheafification is needed at all: $\Gr_G$ is already the presheaf quotient $LG/L^+G$, which seems new already in the classical case of $\GL_n$ over $\bC$. For totally isotropic $G$, we also show that the affine Grassmannian may be formed using polynomial loops. We deduce all of these results from the study of $G$-torsors on $\bP^1_A$ that is ultimately built on the geometry of $\Bun_G$.
\end{abstract}

\begin{altabstract}
Pour un groupe rÃ©ductif $G$ sur un anneau $A$, sa grassmannienne affine $\mathrm{Gr}_G$ joue un rÃ´le important dans un grand nombre de sujets et est typiquement dÃ©finie comme la faisceautisation Ã©tale du quotient de prÃ©faisceaux $LG/L^+G$ du groupe de lacets $LG$ par son sous-groupe d'arcs $L^+G$. Nous montrons que la faisceautisation de Zariski donne le mÃªme rÃ©sultat. De plus, pour $G$ totalement isotrope (par exemple, pour $G$ quasi-dÃ©ployÃ©), nous montrons qu'aucune faisceautisation n'est nÃ©cessaire : $\mathrm{Gr}_G$ est dÃ©jÃ  le quotient de prÃ©faisceaux $LG/L^+G$, ce qui semble nouveau dÃ©jÃ  dans le cas classique d'un $G$ sur $\mathbb{C}$. Pour $G$ totalement isotrope, nous montrons aussi que sa grassmannienne affine peut Ãªtre formÃ©e en utilisant des lacets polynomiaux. Nous dÃ©duisons tous ces rÃ©sultats de l'Ã©tude des torseurs de $G$ sur $\mathbb{P}^1_A$ basÃ© sur la gÃ©omÃ©trie de $\mathrm{Bun}_G$.
\end{altabstract}

\begin{document}

\maketitle

The affine Grassmannian $\Gr_G$ of a reductive group $G$ was introduced by Beilinson--Drinfeld in~\cite[Section~4.5]{BD19} and is instrumental in the geometric Langlands program and other subjects that study $G$-torsors and their moduli. The goal of this article is to show that $\Gr_G$ admits a simpler definition than previously thought: for most $G$, it is simply the presheaf quotient $LG/L^+G$ of the loop group $LG$ by its positive loop subgroup $L^+G$, so that the fpqc or \'etale sheafifications of this quotient that were used previously are not needed, see Theorems~\ref{thm:Zariski} and~\ref{thm:main} for precise statements.

\begin{swapenonce}{Conventions}\label{pp:conv}
As in~\cite[expos\'e~XIX, d\'efinition~2.7]{SGA3IIInew}, a \emph{reductive group scheme} over a scheme $S$ is a smooth, affine $S$-group scheme whose geometric $S$-fibers are connected reductive groups. An $S$-scheme is \emph{locally constant} if fpqc locally on $S$ it becomes isomorphic to some $\bigsqcup_{i \in I} S$.
\end{swapenonce}

\section{The affine Grassmannian and its modular description}

In this section, we fix a base ring $A$ and aim to review the definition and the modular interpretation of the affine Grassmannian $\Gr_\cG$ for a smooth, quasi-affine $A[t]$-group scheme $\cG$, see \propref{prop:modular}.

\begin{swapenonce}{The loop functor}\label{pp:loops}
For a functor $\cX$ on the category of $A\pparens{t}$-algebras (resp.,~$A[t]$-algebras), its \emph{loop functor} $L\cX$ (resp., \emph{positive loop functor} $L^+\cX$) is defined on the category of $A$-algebras~by
\[
L\cX\colon B \mapsto \cX\parens[\big]{B\pparens{t}} \quad\text{(resp., $\ L^+\cX\colon B \mapsto \cX\parens[\big]{B[t]}$)}.
\]

The morphism $L^+\cX \to L\cX$ is often an inclusion, for instance, this is so whenever $\cX$ is a subfunctor of a separated $A[t]$-scheme (see~\cite[corollaire~9.5.6]{EGAI}).

If $\cX$ is an $A[t]$-scheme, then, by~\cite[Theorem~4.1 and~Remark~4.6]{Bha16}, $L^+\cX$ is an $A$-scheme because
\[
\textstyle L^+\cX \cong \varprojlim_{n > 0} \Res_{\parens{A[t]/(t^n)}/A} (\cX_{A[t]/(t^n)}), \quad\text{equivalently,} \quad (L^+\cX)(B) = \varprojlim_{n > 0} \cX\parens[\big]{B[t]/(t^n)}
\]
for every $A$-algebra $B$, compare with~\cite[Chapter~3, Corollary~3.3.7(b)]{CLNS18}. Similarly, if $\cX$ is a quasi-compact and quasi-separated $A[t]$-algebraic space, then, by~\cite[Theorem~4.1]{Bha16} and~\cite[Proposition~\href{https://stacks.math.columbia.edu/tag/05YF}{05YF} and Lemmas~\href{https://stacks.math.columbia.edu/tag/07SF}{07SF} and~\href{https://stacks.math.columbia.edu/tag/05YD}{05YD}]{SP}, $L^+\cX$ is a quasi-compact and quasi-separated $A$-algebraic space. If $\cX$ is even an affine $A[t]$-scheme, then, by considering a presentation of its coordinate ring in terms of generators and relations, $L^+\cX$ is an affine $A$-scheme and $L\cX$ is an ind-affine $A$-ind-scheme, more precisely, there are affine $A$-schemes $X_n$ and closed immersions
\[
\textstyle X_0 \hra X_1 \hra \dotsb \quad\text{such that} \quad L^+\cX = X_0 \quad\text{and} \quad L\cX = \bigcup_{n \ge 0} X_n \quad\text{as functors on $A$-algebras $B$.}
\]
For an $A$-algebra $B$, let $B\{t\}$ be the Henselization of $B[t]$ with respect to the ideal $tB[t]$, see~\cite[Section~2.1.2]{Hitchin-torsors} or~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/0A02}{0A02}]{SP}. It is useful to consider Henselian (resp.,~algebraic; resp.,~polynomial) variants $L_{\Hensel} \cX$ and $L^+_{\Hensel}\cX$ (resp.,~$L_\alg \cX$ and $L^+_\alg\cX$; resp.,~$L_{\mathrm{poly}} \cX$ and $L^+_{\mathrm{poly}}\cX$):
\[
\begin{aligned}
L_{\Hensel} \cX\colon B \mapsto \cX\parens[\big]{B\{ t \}\bracks[\big]{\tfrac{1}{t}}}	 					\quad	& \text{and} \quad L^+_{\Hensel}\cX \colon B \mapsto \cX\parens[\big]{B\{ t \}}, \\
L_\alg\cX \colon B \mapsto \cX\parens[\big]{\parens[\big]{B[t]_{1 + tB[t]}}\bracks[\big]{\tfrac{1}{t}}} 	\quad	& \text{and} \quad L^+_\alg\cX \colon B \mapsto \cX\parens[\big]{B[t]_{1 + tB[t]}}, \\
L_{\mathrm{poly}}\cX \colon B \mapsto \cX\parens[\big]{B[t, t^{-1} ]} 									\quad	& \text{and} \quad L^+_{\mathrm{poly}}\cX \colon B \mapsto \cX\parens[\big]{B[t]},
\end{aligned}
\]
granted that $\cX$ begins life over $A\{t\}\bracks[\big]{\tfrac{1}{t}}$ (resp.,~over $\parens[\big]{A[t]_{1 + tA[t]}}\bracks[\big]{\tfrac{1}{t}}$; resp.,~over $A[t, t^{-1} ]$) and, for $L^+_*\cX$, even already over $A\{t\}$ (resp.,~over $A[t]_{1 + tA[t]}$; resp.,~over $A[t]$). These variant functors are sometimes easier to handle, for instance, they all commute with filtered direct limits in $B$ granted that so does $\cX$, moreover, $L^+_{\mathrm{poly}}\cX$ and $L_{\mathrm{poly}}\cX$ are nothing else but restrictions of scalars.

By the following proposition, for many $\cX$, the functors $L_*^{(+)}\cX$ cannot be improved by sheafifying.
\end{swapenonce}

\begin{prop} \label{prop:L-sheaf}
For a scheme $\cX$ over $A\pparens{t}$, or over $A\{t\}\bracks[\big]{\tfrac{1}{t}}$, or over $\parens[\big]{A[t]_{1 + tA[t]}}\bracks[\big]{\tfrac{1}{t}}$, or over $A[t, t^{-1} ]$ (resp.,~over $A[t]$, or over $A\{t\}$, or over $A[t]_{1 + tA[t]}$, or over $A[t]$) as in Section~\ref{pp:loops} such that every quasi-compact open of $\cX$ is quasi-affine, the functor $L_*\cX$ (resp.,~and also its subfunctor $L^+_*\cX$) is an fpqc sheaf on the category of $A$-algebras.
\end{prop}

\begin{proof} Since $\cX$ is separated, \cite[corollaire~9.5.6]{EGAI} ensures that the map $L^+_*\cX \to L_*\cX$ is indeed an inclusion. Thus, all we need to show is that for an fpqc cover $B \to B'$ of $A$-algebras, the sequence
%\begin{gather*}
%(L_*\cX)(B) \to (L_*\cX)(B') \longrightrightarrows (L_*\cX)(B' \tensor_B B') \\
%\text{(resp., $(L_*^+\cX)(B) \to (L_*^+\cX)(B') \longrightrightarrows (L_*^+\cX)(B' \tensor_B B')$)}
%\end{gather*}
\begin{align*}
					& (L_*\cX)(B) \to (L_*\cX)(B') \longrightrightarrows (L_*\cX)(B' \tensor_B B') \\
(\text{resp.,} \ 	& (L_*^+\cX)(B) \to (L_*^+\cX)(B') \longrightrightarrows (L_*^+\cX)(B' \tensor_B B'))
\end{align*}
is exact. The case of the polynomial variant follows from fpqc descent~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/023Q}{023Q}]{SP} because
\[
B[t] \to B'[t] \longrightrightarrows (B' \tensor_B B')[t] \qquad\text{and}\qquad B[t, t^{-1} ] \to B'[t, t^{-1} ] \longrightrightarrows (B' \tensor_B B')[t, t^{-1} ]
\]
are both fpqc cover sequences. For other variants, since each ring-valued point of $\cX$ factors through a quasi-compact open, we may assume that $\cX$ is quasi-affine. The maps
\[
B\{t\} \to B'\{t\} \qquad\text{and}\qquad B[t]_{1 + tB[t]} \to B'[t]_{1 + tB'[t]}
\]
are both faithfully flat, in particular, they induce surjections on spectra, so the Henselian and algebraic variants reduce further to affine $\cX$. In the power series case, the reduction to affine $\cX$ is more subtle if $B$ and $B'$ are not Noetherian (because then we do not know whether the map $\Spec B'\pparens{t} \to \Spec B \pparens{t}$ is surjective) and will use ideas from~\cite[Lemma~2.2.9(i)]{Hitchin-torsors} as follows.

By realizing $\cX$ as the complement of the vanishing locus of a finitely generated ideal in an affine $A\pparens{t}$-scheme (resp.,~$A[t]$-scheme), to reduce to affine $\cX$ we need to show that elements $b_1, \dotsc, b_n$ in $B\pparens{t}$ (resp.,~in $B[t]$) generate the unit ideal as soon as their images do so in $B'\pparens{t}$ (resp.,~in $B'[t]$). The case of $B[t]$ follows by checking modulo $t$ and using the faithful flatness of $B \to B'$. To treat $B\pparens{t}$, we may assume that $b_1, \dotsc, b_n \in B[t]$ and use the faithfully flat cover \smash{$B[t] \to \parens[\big]{B[t] \tensor_B B'}^h_{(t)}$} (Henselization with respect to the ideal generated by $t$) to reduce to showing that $b_1, \dotsc, b_n$ generate the unit ideal in $\parens[\big]{B[t] \tensor_B B'}^h_{(t)}\bracks[\big]{\tfrac{1}{t}}$. The $t$-adic completion of $\parens[\big]{B[t] \tensor_B B'}^h_{(t)}$ is $B'[t]$ and, by assumption, there are $a_1, \dotsc, a_n \in B'[t]$ with $a_1b_1 + \cdots + a_nb_n = t^N$ in $B'[t]$ for some $N > 0$. By approximating modulo $t^{N + 1}$, therefore, there are $a'_1, \dotsc, a'_n \in \parens[\big]{B[t] \tensor_B B'}^h_{(t)}$ such that
\[
a_1'b_1 + \cdots + a_n'b_n \in t^N + t^{N + 1}\parens[\big]{B[t] \tensor_B B'}^h_{(t)}.
\]
This means that $b_1, \dotsc, b_n$ indeed generate the unit ideal in $\parens[\big]{B[t] \tensor_B B'}^h_{(t)}\bracks[\big]{\tfrac{1}{t}}$, as desired.

Now that $\cX$ is affine, the functor $\cX(-)$ turns fiber products of rings into fiber products of sets. Thus, all we need to show is the exactness of the horizontal sequences in the commutative diagram
\[
\xymatrix@R=12pt{ B[t]_{1 + tB[t]} \ar@{^(->}[d] \ar[r] & B'[t]_{1 + tB'[t]} \ar@{^(->}[d]\ar@<0.5ex>[r]\ar@<-0.5ex>[r] & (B' \tensor_B B')[t]_{1 + t(B' \tensor_B B')[t]} \ar@{^(->}[d] \\
B\{t\} \ar[r]\ar@{^(->}[d] &B'\{t\} \ar@<0.5ex>[r]\ar@<-0.5ex>[r]\ar@{^(->}[d] &(B' \tensor_B B')\{t\}\ar@{^(->}[d] \\
B[t] \ar[r] &B'[t] \ar@<0.5ex>[r]\ar@<-0.5ex>[r] &(B' \tensor_B B')[t], }
\]
which implies the corresponding exactness after further inverting $t$; here the bottom vertical maps are injective by~\cite[Section~2.1.2]{Hitchin-torsors} (a limit argument to reduce to finite type $\bZ$-algebras). The exactness of the bottom row is seen coefficientwise. Thus, by fpqc descent, it is enough to show that
\[
\begin{aligned} B'\{t\} \tensor_{B\{t\}} B'\{t\} 					& \longhookrightarrow (B' \tensor_B B')\{ t\}, \\
B'[t]_{1 + tB'[t]} \tensor_{B[t]_{1 + tB[t]}} B'[t]_{1 + tB'[t]} 	& \longhookrightarrow (B' \tensor_B B')[t]_{1 + t(B' \tensor_B B')[t]}.
\end{aligned}
\]
Since every \'{e}tale $(B'\tensor_B B')[t]$-algebra $(B'\tensor_B B')$-fiberwise has no embedded associated primes, both of these injections follow from~\cite[premiÃ¨re~partie, corollaire~3.2.6]{RG71} and a limit argument.
\end{proof}

\begin{rema} \label{rem:ind-quasi-affine}
In \propref{prop:L-sheaf}, the assumption on quasi-compact opens holds if $\cX$ is a subscheme of an affine scheme. In general, a scheme whose quasi-compact opens are all quasi-affine is \emph{ind-quasi-affine}, see~\cite[Definition~2.2.5]{Hitchin-torsors} and the references to~\cite{SP} given there for details. We avoid this terminology here because it may be confusing in the context of ind-schemes. The automorphism scheme of a reductive group is such an $\cX$ that need not be quasi-affine, see~\cite[Section~1.3.7]{torsors-regular}. Another useful example is locally constant schemes $\cX$, for which even $L\cX = L^+\cX$ as follows.
\end{rema}

\begin{coro}\label{cor:L-Lplus}
Let $A$ be a ring.
\begin{enumerate}
\item \label{m:LL-a} In the following diagram, the squares are Cartesian and the maps are bijective on idempotents:
\[
\q\xymatrix@C=15pt@R=15pt{ A \ar@{^(->}[r]\ar@{_(->}[rd] &A[t] \ar@{^(->}[r] \ar@{^(->}[d] & A[t]_{1 + tA[t]} \ar@{^(->}[r]\ar@{^(->}[d] & A\{t \} \ar@{^(->}[r]\ar@{^(->}[d] & A[t] \ar@{^(->}[d] \\
& A[t, t^{-1} ] \ar@{^(->}[r]& \parens[\big]{A[t]_{1 + tA[t]}}\bracks[\big]{\tfrac{1}{t}} \ar@{^(->}[r] &A\{t \}\bracks[\big]{\tfrac{1}{t}} \ar@{^(->}[r] & A\pparens{t}. }
\]
\item \label{m:LL-b} For every scheme $\cX$ over $A[t]$, or over $A\{t\}$, or over $A[t]_{1 + tA[t]}$, or over $A[t]$ as in Section~\ref{pp:loops} such that $\cX$ is locally constant (see~Section~\ref{pp:conv}), we have $L_*^+\cX \isomto L_*\cX$.
\end{enumerate}
\end{coro}

\begin{proof}
\mbox{}
\begin{enumerate}
\item[\eqref{m:LL-a}] As indicated, the maps are all injective, either by inspection, or by faithful flatness, or by a limit argument given in~\cite[Section~2.1.2]{Hitchin-torsors}. The squares are Cartesian by~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/0BNR}{0BNR}]{SP}. As for idempotents, since the maps are injective, it suffices to recall from~\cite[Corollary~2.1.19]{Hitchin-torsors} that the map $A \to A\pparens{t}$ is bijective on idempotents (as one may also show directly).
\item[\eqref{m:LL-b}] By~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/0AP8}{0AP8}]{SP}, we may check fpqc locally on the base that our locally constant $\cX$ satisfies the assumptions of \propref{prop:L-sheaf}, more precisely, that every quasi-compact open of $\cX$ is quasi-affine. In particular, in the case when $\cX$ begins life over $A\{t\}$, the square
\[
\qq\xymatrix{
\cX\parens[\big]{A\{t\}} \ar@{^(->}[r]\ar[d] & \cX\parens[\big]{A\{t\}\bracks[\big]{\tfrac{1}{t}}} \ar[d] \\
\cX\parens[\big]{A[t]} \ar@{^(->}[r] & \cX\parens[\big]{A\pparens{t}} }
\]
is Cartesian by~\cite[Proposition~2.2.12]{Hitchin-torsors}, and likewise with $A[t]_{1 + tA[t]}$ or $A[t]$ in place of $A\{t\}$. Since we may vary $A$, this reduces us to showing that the inclusion $\cX\parens[\big]{A[t]} \subset \cX\parens[\big]{A\pparens{t}}$ is an equality. For this, since loc.~cit.~also implies that for every prime $\fp \subset A$ the square
\[
\qq\xymatrix{
\cX\parens[\big]{A[t] \tensor_A A_\fp} \ar@{^(->}[r]\ar[d] & \cX\parens[\big]{A\pparens{t} \tensor_A A_\fp} \ar[d] \\
\cX\parens[\big]{A_\fp[t]} \ar@{^(->}[r] & \cX\parens[\big]{A_\fp\pparens{t}} }
\]
is Cartesian, by combining this square with spreading out, we may assume further that $A$ is local. At this point, \propref{prop:L-sheaf} allows us to replace $A$ by its strict Henselization.

Once $A$ is strictly Henselian local, we fix any $A\pparens{t}$-point of $\cX$ and form its schematic image to get a closed subscheme $Z \subset \cX$, see~\cite[Definition~\href{https://stacks.math.columbia.edu/tag/01R7}{01R7}]{SP}. It is enough to show that
\[
\qq Z \overset{?}{\cong} \Spec\parens[\big]{A [t]}.
\]
We first claim that the map $Z \to \Spec\parens[\big]{A [t]}$ is surjective. For this, since $\cX$ is locally constant, we may choose a faithfully flat $A [t]$-algebra $B$ such that $\cX_B \cong \bigsqcup_{i \in I} \Spec B$. By~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/081I}{081I}]{SP}, the formation of $Z$ commutes with base change to $B$, so we need to show that the schematic image of any $B\bracks[\big]{\tfrac{1}{t}}$-point of $\cX_B$ surjects onto $\Spec B$. It does: otherwise, by the explicit nature of $\cX_B$, the open immersion $\Spec \parens[\big]{B\bracks[\big]{\tfrac{1}{t}}} \subset \Spec B$ would factor through some proper closed subscheme, contradicting its schematic density.

Since $A[t]$ is strictly Henselian local and $\cX$ is \'etale, \cite[th\'eor\`eme~18.5.11]{EGAIV4} ensures that for every $z \in Z$ above the maximal ideal of $A[t]$, we have
\[
\qq \cX \cong \Spec (\sO_{\cX,\, z}) \sqcup \cX' \quad\text{with} \ \sO_{\cX,\, z} \cong A[t].
\]
We have already argued that such a $z$ exists, and~\eqref{m:LL-a} shows that $\Spec\parens[\big]{A\pparens{t}}$ inherits connectedness from $\Spec A$, and so ensures that $Z \cap \cX' = \emptyset$. The desired $Z \cong \Spec\parens[\big]{A [t]}$ now follows from the schematic density of the open immersion
\[
\Spec\parens[\big]{A\pparens{t}} \subset \Spec\parens[\big]{A[t]}. \qedhere
\]
\end{enumerate}
\end{proof}

\begin{swapenonce}{The affine Grassmannian}\label{pp:GrG}
In the setting of~Section~\ref{pp:loops}, if the functor $\cX$ is group valued, then so are the functors $L^+_*\cX$ and $L_*\cX$. With this in mind, for a smooth, quasi-affine $A[t]$-group scheme $\cG$, its \emph{affine Grassmannian} is the pointed set valued functor defined by
\begin{equation}\label{eqn:GrG}
\Gr_\cG \coloneqq \parens[\big]{L\cG/L^+\cG}_\et,
\end{equation}
where, as indicated, the sheafification of the presheaf quotient $L\cG/L^+\cG$ is formed in the \'etale topology.

If instead $\cG$ begins life as a smooth, quasi-affine $A\{t\}$-group scheme, then~\cite[Example~2.2.19]{Hitchin-torsors} (which is based on approximation and algebraization techniques) ensures that we may use Henselian loops instead, more precisely, that we have an identification of presheaf quotients
\[
L\cG/L^+\cG \cong L_{\Hensel}\cG/L_{\Hensel}^+\cG.
\]
In many situations, the affine Grassmannian $\Gr_\cG$ is represented by an $A$-ind-scheme, which is often even ind-projective over $A$, see, for instance, \cite{PR08}. This proceeds by embedding $\cG$ into some $\GL_n$, which is not always possible with our general assumptions, so we will not use representability results.
\end{swapenonce}

In the literature one sometimes finds the affine Grassmannian defined as the fpqc sheafification $\parens[\big]{L\cG/L^+\cG}_{\fpqc}$. A priori this makes no mathematical sense: even for a field, isomorphism classes of its fpqc covers form a proper class, so the fpqc sheafification may not exist; however, by the following proposition, this ``fpqc sheafification approach'' gives the same result for our $\cG$ as above.

\begin{prop} \label{prop:modular}
For a ring $A$ and a smooth, quasi-affine $A[t]$-group scheme $\cG$, the affine Grassmannian $\Gr_\cG$ is an fpqc sheaf and has the following modular description on the category of $A$-algebras~$B$:
\[
\Gr_\cG(B) \cong
\left\{ (\cE, \iota) \;\middle\vert\;
\begin{aligned}
& \text{\normalfont $\cE$ is a $\cG$-torsor over $B[t]$}\\
& \text{\normalfont $\iota \in \cE\parens[\big]{B\pparens{t}}$ is its trivialization over $B\pparens{t}$}
\end{aligned}
\right\} /\sim,
\]
while the presheaf quotient $L\cG/L^+\cG$ is a subfunctor of\/ $\Gr_\cG$ parametrizing those $(\cE, \iota)$ with $\cE$ trivial.
\end{prop}

\begin{proof} For the trivial $\cG$-torsor, $L^+\cG$ parametrizes its $\cG$-torsor automorphisms over $(-)[t]$ while $L\cG$ parametrizes its trivializations $\iota$ over $(-)\pparens{t}$, so the presheaf quotient $L\cG/L^+\cG$ is the claimed subfunctor of $\Gr_\cG$, granted that the latter has the displayed modular description. Moreover, by the smoothness and quasi-affineness of $\cE$ inherited from $\cG$ (see~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/0247}{0247}]{SP}) and by the infinitesimal lifting of sections (compare also with~\cite[Proposition~2.1.4]{Hitchin-torsors}), each $(\cE, \iota)$ lands in this subfunctor Ã©tale locally on $B$, so the sought modular description will follow once we argue that it defines an fpqc sheaf. For this, since the pairs $(\cE, \iota)$ have no nontrivial automorphisms, all that remains is to show that the groupoid-valued~functor
\[
B \mapsto
\left\{ (\cE, \iota) \;\middle\vert\;
\begin{aligned}
& \text{$\cE$ is a $\cG$-torsor over $B[t]$}\\
& \text{$\iota \in \cE\parens[\big]{B\pparens{t}}$ is its trivialization over $B\pparens{t}$}
\end{aligned}
\right\}
\]
is a stack for the fpqc topology on $A$-algebras $B$. For this, let $B \to B'$ be an fpqc cover of $A$-algebras and let $(\cE', \iota')$ be a pair for $B'$ equipped with a descent datum with respect to this cover. It suffices to uniquely descend $\cE'$ to a $\cG$-torsor $\cE$ over $B[t]$: the trivialization $\iota'$ will then also descend because $L\cE$ is an fpqc sheaf by \propref{prop:L-sheaf}. The usual fpqc descent for $\cG$-torsors (see~\cite[Lemma~\href{https://stacks.math.columbia.edu/tag/0247}{0247}]{SP}) gives us a unique compatible system of $\cG$-torsor descents $\cE_n$ over $B[t]/(t^{n+1})$ for $n \ge 0$. It then remains to uniquely algebraize this sequence to a $\cG$-torsor $\cE$ over $B[t]$, and there are several ways to do this. Perhaps the most direct is to apply the algebraization result~\cite[Theorem~8.1]{BHL17} to the classifying stack $\bbB \cG$. A somewhat more elementary approach is to first use~\cite[Theorem~2.1.6]{Hitchin-torsors} to lift $\cE_0$ to a $\cG$-torsor $\cE$ over $B[t]$ and to then inductively build a compatible sequence of isomorphisms
\[
\restr{\cE}{B[t]/(t^{n + 1})} \simeq \cE_n,
\]
by using loc.~cit.~and the smoothness of the quasi-affine $B[t]$-group $\underline{\Aut}_\cG(\cE)$.
\end{proof}

\section{The affine Grassmannian as a Zariski quotient}

Even though the affine Grassmannian of $\cG$ is defined as the \'etale sheafification of the presheaf quotient $L\cG/L\cG^+$ (see~\eqref{eqn:GrG}), in \theoref{thm:Zariski} below we show that the Zariski sheafification suffices when $\cG$ is reductive and descends to $A$. For intuition for why the Zariski topology may be enough, we recall that the inclusion $L^+\cG \subset L\cG$ is morally similar to the inclusion $P \subset G$ of a parabolic subgroup of a reductive group scheme, and that $(G/P)_\fpqc \cong (G/P)_\Zar$ because over any semilocal ring parabolic subgroups of the same type are conjugate, see~\cite[exposÃ©~XXVI, corollaire~5.2]{SGA3IIInew}.

We will deduce that the Zariski sheafification is enough from the following result about torsors over $\bP^1_A$ that is proved by studying the geometry of the algebraic stack $\Bun_G$ that parametrizes such torsors, or in~\cite{PS23a} and~\cite{PS23b} by a different approach. Various weaker and more technical earlier variants of this result would suffice as well, for instance, \cite[Theorem~6]{Fed22a} or~\cite[Proposition~5.3.6]{torsors-regular}. In effect, in some sense, we obtain the main results of this article by ascending geometric information along the uniformization map $\Gr_G \to \Bun_G$.

\begin{theo}[{\cite[Theorem~3.6]{totally-isotropic}}]\label{thm:CF-input}
Let $G$ be a reductive group scheme over a semilocal ring $A$. Every $G$-torsor $E$ over\/ $\bP^1_A$ is $A$-sectionwise constant, equivalently, $\restr{E}{\{ t= 0 \}} \simeq \restr{E}{\{ t= \infty \}}$.
\end{theo}

We will use \theoref{thm:CF-input} through its following consequence for reductive group torsors over $A[t]$.

\begin{prop} \label{prop:semilocally}
Let $A$ be a semilocal ring and let $\cG$ be an $A[t]$-group scheme that is an extension of an $A[t]$-group $\cG^\et$ that is locally constant (see Section~\ref{pp:conv}) by a reductive $A[t]$-group scheme $\cG^0$ such that $\cG^0_{A\pparens{t}}$ descends to a reductive $A$-group scheme. No nontrivial $\cG$-torsor over $A[t]$ trivializes over $A\pparens{t}$, that is, we have
\[
\Ker\parens[\bigg]{H^1\parens[\big]{A[t], \cG} \ra H^1\parens[\big]{A\pparens{t}, \cG}} = \{ *\};
\]
in particular, $\cG^0$ itself descends to a reductive $A$-group scheme.
\end{prop}

\begin{proof} Let $\cE$ be a $\cG$-torsor over $A[t]$ that trivializes over $A\pparens{t}$. By \cororef{cor:L-Lplus}\eqref{m:LL-b}, every $A\pparens{t}$-point $e \in \cE\parens[\big]{A\pparens{t}}$ gives rise to an $A\pparens{t}$-point of the $\cG^\et$-torsor $\cE/\cG^0$ that extends uniquely to an $A[t]$-point $\bar{e} \in (\cE/\cG^0)\parens[\big]{A[t]}$. The preimage of $\bar{e}$ in $\cE$ is a $\cG^0$-torsor that trivializes over $A\pparens{t}$. We may replace $\cE$ by this preimage to reduce to the case when $\cG = \cG^0$.

Now that $\cG = \cG^0$, suppose first that $\cG$ descends to a reductive $A$-group $G$. Due to the triviality over $A\pparens{t}$, we may patch $\cE$ with the trivial $G$-torsor over $\bP^1_A \setminus \{ t = 0\}$ (see, for instance, \cite[Lemma~2.2.11(b)]{Hitchin-torsors}) to build a $G$-torsor $E$ over $\bP^1_A$ such that $\restr{E}{\{t = 0\}} \simeq \restr{\cE}{\{t = 0\}}$ and $\restr{E}{\{t = \infty\}}$ is trivial. By \theoref{thm:CF-input}, then $\restr{\cE}{\{t = 0\}}$ is also trivial, to the effect that, by the infinitesimal lifting of sections due to smoothness, $\cE$ is trivial, too, as desired (compare also with~\cite[Proposition~2.1.4]{Hitchin-torsors}).

To complete the proof, it remains to show that the reductive $A$-group scheme $G$ for which we have $G_{A\pparens{t}} \simeq \cG_{A\pparens{t}}$ also descends $\cG$, that is, that already $G_{A[t]} \simeq \cG_{A[t]}$. By \cororef{cor:L-Lplus}\eqref{m:LL-a} and the classification of reductive group schemes (recalled in~\cite[Sections~1.3.1 and~1.3.7]{torsors-regular}), our $\cG$ corresponds to an \smash{$\underline{\Aut}_\gp(G)$}-torsor over $A[t]$ that trivializes over $A\pparens{t}$. However, as recalled in loc.~cit.,~$\underline{\Aut}_\gp(G)$ is an extension of a locally constant $A$-group by the reductive $A$-group $G^\ad$. In particular, the case settled in the first two paragraphs of the proof applies and shows that the $\underline{\Aut}_\gp(G)$-torsor in question is trivial already over $A[t]$, so that $G_{A[t]}\simeq \cG$, as desired.
\end{proof}

\begin{coro} \label{cor:H1-local}
For a reductive group scheme $G$ over a semilocal ring $A$, no nontrivial $G$-torsor trivializes over $A\pparens{t}$, in other words, we have
\[
\Ker\parens[\bigg]{H^1(A, G) \ra H^1\parens[\big]{A\pparens{t}, G}} = \{ *\}.
\]
\end{coro}

\begin{proof} In view of \propref{prop:semilocally}, it suffices to recall from~\cite[Theorem~2.1.6]{Hitchin-torsors} that
\[
H^1(A, G) \isomto H^1\parens[\big]{A[t], G}. \qedhere
\]
\end{proof}

\begin{rema} The results of~\cite{FG21} suggest that the reductivity assumption may be nonessential for \cororef{cor:H1-local}. It would therefore be interesting to find a more general result of this type.
\end{rema}

We are ready for the promised sufficiency of the Zariski sheafification for $\Gr_\cG$ in the case when $\cG$ is (constant) reductive. One may compare this to its earlier variant~\cite[Proposition~14]{Bac19} that restricted to smooth $A$ over a field and deduced the conclusion from the Grothendieck--Serre conjecture. For us, it is the study of the latter that has indirectly led to the results of this article.

\begin{theo} \label{thm:Zariski}
For a ring $A$ and an $A[t]$-group scheme $\cG$ that is an extension of a finite \'etale $A[t]$-group $\cG^\et$ by a reductive $A[t]$-group scheme $\cG^0$ that descends to a reductive $A$-group scheme, the affine Grassmannian $\Gr_\cG$ is the Zariski sheafification of the presheaf quotient $L\cG/L^+\cG$, that is,
\[
\Gr_\cG \cong\parens[\big]{L\cG/L^+\cG}_\Zar.
\]
\end{theo}

\begin{proof} By the modular interpretation supplied by \propref{prop:modular}, the possibility of varying $A$, and the infinitesimal lifting of sections due to smoothness, all we need to show is that for any $\cG$-torsor $\cE$ over $A[t]$ that trivializes over $A\pparens{t}$, the $\cG$-torsor $\restr{\cE}{\{t = 0\}}$ trivializes Zariski locally on $A$. However, \propref{prop:semilocally} ensures that $\restr{\cE}{\{t = 0\}}$ trivializes even Zariski semilocally on $A$.
\end{proof}

\begin{remas}
\mbox{}
\begin{enumerate}
\item We do not know whether the reductivity assumption is critical for \theoref{thm:Zariski}, for instance, whether the Zariski sheafification also suffices for parahoric groups. Certainly, it does in the case when $\cG$ is a reductive $A[t]$-group scheme and $\cP$ is a smooth, quasi-affine $A[t]$-group scheme equipped with an $A[t]$-morphism $\cP \to \cG$ that modulo $t$ reduces to an inclusion of a parabolic subgroup: indeed, by~\cite[Theorem~2.1.6]{Hitchin-torsors} and~\cite[(1.3.5.2)]{torsors-regular}, these assumptions ensure that $H^1\parens[\big]{B[t], \cP} \subset H^1\parens[\big]{B [t], \cG}$ for any semilocal $A$-algebra $B$, so
\[
\qqq\Ker\parens[\bigg]{H^1\parens[\big]{B [t], \cG} \ra H^1\parens[\big]{B\pparens{t}, \cG} } = \{ *\} \ \Longrightarrow \ \Ker\parens[\bigg]{H^1\parens[\big]{B[t], \cP} \ra H^1\parens[\big]{B\pparens{t}, \cP} } = \{ *\},
\]
to the effect that \propref{prop:modular} and \theoref{thm:Zariski} imply the claimed $\Gr_\cP \cong \parens[\big]{L\cP/L^+\cP}_\Zar$.
\item We do not know whether \theoref{thm:Zariski} admits a version for the Witt vector affine Grassmannian. For a version of \theoref{thm:Zariski} for the $B_\dR^+$-affine Grassmannian, see~\cite[Theorem~3.1]{BdR-sheafification}.
\end{enumerate}
\end{remas}

\section{The affine Grassmannian as a presheaf quotient}

For most reductive group schemes $\cG$, even the Zariski sheafification is not needed when forming the affine Grassmannian $\Gr_\cG$: in \theoref{thm:main} below, we show that the latter often agrees already with the presheaf quotient $L\cG/L^+\cG$. This appears to be new already for $\GL_{n}$ over $\bC$ and is based on the finer variant of \theoref{thm:CF-input} recorded in \theoref{thm:CF2} below, which, in addition to the geometry of $\Bun_G$, uses Quillen patching for torsors over $\bA^1_A$ to progress beyond semilocal $A$. For this variant, the relevant condition on $\cG$ is the following.

\begin{defi}[{\cite[Definition~8.1]{split-unramified}}] A semisimple group scheme $G$ over a scheme $S$ is \emph{totally isotropic} if in the canonical decomposition of~\cite[exposÃ©~XXIV, proposition~5.10(i)]{SGA3IIInew}:
\[
\textstyle G^\ad \cong \prod_{i \in \{ A_n, B_n, \dotsc, G_2 \}} \Res_{S_i/S}(G_i)
\]
of its adjoint quotient $G^\ad$, where $i$ ranges over the types of connected Dynkin diagrams, $S_i$ is a finite \'etale $S$-scheme, and $G_i$ is an adjoint semisimple $S_i$-group with simple geometric fibers of type $i$, Zariski locally on $S$ each $G_i$ has a parabolic $S_i$-subgroup that contains no $S_i$-fiber of\/ $G_i$.
\end{defi}

\begin{exem}
Slightly informally, $G$ is totally isotropic if Zariski locally on $S$ it has a parabolic subgroup containing no factor of the adjoint group $G^\ad$. To see this, recall that parabolic subgroups of $G$ correspond to those of $G^\ad$, which correspond to collections of parabolic subgroups of $G_i$, one for each $i$, see~\cite[Section~1.3.4]{torsors-regular}. Certainly, every quasi-split (so also every split) semisimple group is totally isotropic.
\end{exem}

\begin{theo}[{\cite[Theorem~4.2]{totally-isotropic}}]\label{thm:CF2}
Let $G$ be a reductive group scheme over a ring $A$ such that $G^{\ad}$ is totally isotropic. For a $G$-torsor $E$ over\/ $\bP^1_A$, if\/ $\restr{E}{\{t = \infty\}}$ is trivial, then $\restr{E}{\bA^1_A}$ is also trivial.
\end{theo}

\begin{theo}\label{thm:main}
For a ring $A$ and an $A[t]$-group scheme $\cG$ that is an extension of a finite \'etale $A[t]$-group $\cG^\et$ by a reductive $A[t]$-group scheme $\cG^0$ that descends to a reductive $A$-group scheme whose adjoint quotient is totally isotropic, the affine Grassmannian\/ $\Gr_\cG$ is the presheaf quotient $L\cG/L^+\cG$, that is,
\[
\Gr_\cG \cong L\cG/L^+\cG.
\]
\end{theo}

\begin{proof} As in the proof of \theoref{thm:Zariski}, by the modular description of \propref{prop:modular} and the possibility of varying $A$, we need to show that no nontrivial $\cG$-torsor $\cE$ over $A[t]$ trivializes over $A\pparens{t}$. For this, as in the proof of \propref{prop:semilocally}, \cororef{cor:L-Lplus}\eqref{m:LL-b} immediately reduces us to the case when $\cG = \cG_0$ and $\cG$ is the base change of a reductive $A$-group scheme $G$ whose adjoint quotient is totally isotropic.

To treat this case, we again patch $\cE$ with the trivial $G$-torsor over $\bP^1_A \setminus \{ t= 0\}$ to build a $G$-torsor $E$ over $\bP^1_A$ such that $E_{A[t]} \simeq \cE$ and $\restr{E}{\{t = \infty\}}$ is trivial. By \theoref{thm:CF2}, this last condition forces $\restr{E}{\bA^1_A}$~to be trivial. However, then $\cE$ is trivial, too, as desired.
\end{proof}

We recall from~\cite[Theorems~2.1.24 and~3.1.7]{Hitchin-torsors} that the following global variant of \cororef{cor:H1-local} was known when $G$ is either a pure inner form of $\GL_n$ or a torus.

\begin{coro} For a reductive group scheme $G$ over a ring $A$ with $G^{\ad}$ totally isotropic, no nontrivial $G$-torsor over $A$ trivializes over $A\pparens{t}$, in other words, we have
\[
\Ker\parens[\bigg]{H^1(A, G) \ra H^1\parens[\big]{A\pparens{t}, G}} = \{ *\}.
\]
\end{coro}

\begin{proof} \propref{prop:modular} and \theoref{thm:main} show that no nontrivial $G$-torsor over $A[t]$ trivializes over $A\pparens{t}$. Thus, it suffices to recall from~\cite[Theorem~2.1.6]{Hitchin-torsors} that
\[
H^1(A, G) \isomto H^1\parens[\big]{A[t], G}. \qedhere
\]
\end{proof}

Finally, we note that the preceding proof shows that the affine Grassmannian in the totally isotropic case may even be formed using polynomial loops as follows.

\begin{theo} \label{thm:alg-loops}
For a reductive group scheme $G$ over a ring $A$ with $G^{\ad}$ totally isotropic, its affine Grassmannian $\Gr_G$ may be formed as the presheaf quotient using the polynomial loops, more precisely,~we~have
\[
\Gr_G \cong L_{\mathrm{poly}} G/L^+_{\mathrm{poly}} G,
\]
explicitly, we have
\[
\begin{split}
G\parens[\big]{A\pparens{t}}/G\parens[\big]{A[t]}		& \cong G\parens[\big]{A\{ t \}\bracks[\big]{\tfrac{1}{t}}}/G\parens[\big]{A\{ t \}} \\
													& \cong G\parens[\big]{\parens[\big]{A[t]_{1 + tA[t]}}\bracks[\big]{\tfrac{1}{t}}}/G\parens[\big]{A[t]_{1 + tA[t]}} \\
													& \cong G\parens[\big]{A[t, t^{-1} ]}/G\parens[\big]{A[t]},
\end{split}
\]
equivalently,
\[
\begin{aligned}
G\parens[\big]{A\pparens{t}}						& = G\parens[\big]{A[t, t^{-1}]}G\parens[\big]{A[t]}, \\
G\parens[\big]{A\{ t \}\bracks[\big]{\tfrac{1}{t}}} 			& = G\parens[\big]{A[t, t^{-1}]}G\parens[\big]{A\{ t \}}, \\
G\parens[\big]{\parens[\big]{A[t]_{1 + tA[t]}}\bracks[\big]{\tfrac{1}{t}}} 	& = G\parens[\big]{A[t, t^{-1}]}G\parens[\big]{A[t]_{1 + tA[t]}}.
\end{aligned}
\]
\end{theo}

These equalities seem elementary, but we do not know how to argue them directly even for $G = \GL_n$.

\begin{proof} Since $G$ is affine, its functor of points preserves the Cartesianness of the squares from \cororef{cor:L-Lplus}\eqref{m:LL-a}. In particular, the map
\[
G\parens[\big]{A[t, t^{-1} ]}/G\parens[\big]{A[t]} \longhookrightarrow G\parens[\big]{A\pparens{t}}/G\parens[\big]{A[t]}
\]
is injective, and so are its counterparts for algebraic or Henselian loops in place of polynomial loops. Thus, all we need to show is that this map is also surjective or, equivalently, that
\[
G\parens[\big]{A[t, t^{-1} ]}\backslash G\parens[\big]{A\pparens{t}}/G\parens[\big]{A[t]} = \{ *\}.
\]
However, if this double quotient was nontrivial, then we could use patching (for instance, \cite[Lemma~2.2.11~(b)]{Hitchin-torsors}) to build a nontrivial $G$-torsor over $\bA^1_A$ that would trivialize both over $\bG_{m,\, A}$ and also over the formal completion along $\{t = 0\}$. We could then extend this $G$-torsor to all of $\bP^1_A$ by patching with the trivial torsor at infinity, and thus obtain a contradiction to \theoref{thm:CF2}.
\end{proof}

\begin{rema}
We do not know the extent to which the assumptions of Theorems~\ref{thm:main} and~\ref{thm:alg-loops} are optimal because they are imposed by our proofs, for instance, we do not know whether the affine Grassmannian $\Gr_G$ agrees with the presheaf quotient $LG/L^+G$ for every (possibly not totally isotropic) reductive $A$-group $G$, although we expect that it does not. On the other hand, since our proofs are reductions to general results about torsors over $\bP^1_A$, the reader will have no trouble adapting them to various close variants of the affine Grassmannian that are sometimes considered in the literature, for instance, to affine Grassmannians constructed using general relative Cartier divisors in $\bA^1_A$ in place of $\{t = 0\}$.
\end{rema}

\section*{Acknowledgments}

This article was written in relation to the Jacques Herbrand prize of the French Academy of Sciences. I thank the Academy for the prize and for the invitation to submit to the \emph{Comptes Rendus de l'Acad\'emie des Sciences} on this occasion. I thank Alexis Bouthier, Ofer Gabber, and Alex Youcis for helpful conversations and correspondence.

\section*{Declaration of interests}

The author does not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and has declared no affiliations other than their research organizations.

\printbibliography

\end{document}
